Author's response to reviews

Title: Socio-demographic factors related to periodontal status and tooth loss of pregnant women in Mbale district, Uganda.

Authors:

   Isaac Okullo (iokullo@med.mak.ac.ug)

Version: 5 Date: 7 July 2009

Author's response to reviews: see over
Dear Editor

Submission of revised manuscript

We are delighted to hear our manuscript will be published in your prestigious journal. We have made the editorial changes as presented below (see attachment) and now submit a final version of “Socio-demographic factors related to periodontal status and tooth loss of pregnant women in Mbale district, Uganda”.

Thank you for your positive consideration.

Yours sincerely

Margaret Wandera.
<table>
<thead>
<tr>
<th></th>
<th>Major Revision Requested</th>
<th></th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>Author groups - If you wish to include the name of the group in the list of authors on the title page, you must remove the names of individual members of the group from the authors list in the submission system and the manuscript. You must also add the name of the group to the first name box in the submission system, and add the abbreviation of the name in brackets to the last name box. All the members of the group should then be listed in the Acknowledgements section. If you do this, you must also include all the authors in the submission system. You should omit the name of the group from the list of authors on the title page and the submission system, but you should add a note in the Acknowledgements listing the members of the group, and describe the role of the group in this study in the authors’ contributions section.</td>
<td>• We have complied with alternative one and removed duplicate names in submission on system. On the title of manuscript we have followed the PROMISE EBF format which was agreed upon by consortium. Please see article from similar group and published below.</td>
</tr>
<tr>
<td></td>
<td>Tables as additional files - We notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.</td>
<td>• Table 2 is additional not to be part of published manuscript and cited in manuscript</td>
</tr>
<tr>
<td>2</td>
<td>Minor revision Requested</td>
<td></td>
</tr>
<tr>
<td></td>
<td>Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.</td>
<td>• Noted.</td>
</tr>
</tbody>
</table>
Research

'No sister, the breast alone is not enough for my baby' a qualitative assessment of potentials and barriers in the promotion of exclusive breastfeeding in southern Zambia

Eli Fjeld*, Seter Siziya2, Mary Katepa-Bwalya2,3, Chipepo Kankasa3, Karen Marie Moland1,4, Thorkild Tylleskär1 for the PROMISE-EBF Study Group5

Address: 1Centre for International Health, University of Bergen, Norway, 2Department of Community Medicine, University of Zambia (UNZA/SOM), PO Box, 50110 Lusaka, Zambia, 3Department of Paediatrics and Child Health, University of Zambia (UNZA/SOM), PO Box, 50110 Lusaka, Zambia, 4Faculty of Health and Social Sciences, Bergen University College, Norway and 5see complete list at end of article.

Email: Eli Fjell* - Eli.Fjell@student.uit.no; Seter Siziya - assamza@yahoo.com; Mary Katepa-Bwalya - chikwa@zamnet.zm; Chipepo Kankasa - chkankasa@zamnet.zm; Karen Marie Moland - Karen.Moland@cbh.uit.no; Thorkild Tylleskär - Thorkild.Tylleskar@cbh.uit.no; the PROMISE-EBF Study Group - Thorkild.Tylleskar@cbh.uit.no

* Corresponding author

Published: 5 November 2008
Received: 2 October 2007
Accepted: 5 November 2008

This article is available from: http://www.internationalbreastfeedingjournal.com/content/3/1/26

© 2008 Fjell et al; licensee BioMed Central Ltd.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/2.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.